# WORKING PAPER SERIES

## UNDERSTANDING ACCESS TO HIGHER EDUCATION AMONGST HUMANITARIAN MIGRANTS: A LONGITUDINAL ANALYSIS OF AUSTRALIAN SURVEY DATA

Francisco (Paco) Perales School of Social Science, The University of Queensland

Ning Xiang Institute for Social Science Research, The University of Queensland

**Lisa Hartley** Centre for Human Rights Education, Curtin University

Matthias Kubler Institute for Social Science Research, The University of Queensland

Wojtek Tomaszewski Institute for Social Science Research, The University of Queensland

A more recent version of this paper was published as Perales F, Xiang N, Hartley L, Kuber M and Tomaszewski W. (2022) Understanding access to higher education amongst humanitarian migrants: an analysis of Australian longitudinal survey data. Higher Education, 84, 373-397. DOI: <u>10.1007/s10734-021-00772-x</u>

No. 2020-31 December 2020











### NON-TECHNICAL SUMMARY

Humanitarian migrants are amongst the most marginalised population groups in countries within the Global North, including Australia. An important channel for these migrants to successfully settle into the host society and improve their socio-economic outcomes is participation in the local education system, particularly in higher-education options. However, we know surprisingly little about the socio-demographic factors that facilitate or constrain access to (higher) education amongst humanitarian migrants, with evidence from robust quantitative studies being particularly scarce. The present study fills this important gap in knowledge by analysing Australian longitudinal survey data (*Building a New Life in Australia*; n=2,109 migrants and 8,668 person-year observations).

Key results indicated that higher English-language proficiency and pre-arrival education levels are core factors fostering greater engagement with the Australian higher-education system amongst humanitarian migrants. For example, humanitarian migrants who entered Australia with no qualifications participated in the education system in 10.1% of the observations, compared to 11.9% for those with some schooling, 14.5% for those with a trade qualification, and 19% for those with a university degree. Further, those who entered Australia with no qualifications selected a higher education course in 4.1% of observations, compared to 9.4% for those with some schooling, 13% for those with a trade qualification, and 21.3% for those with a degree.

Concerning gender differences, humanitarian-migrant women in our sample exhibited a greater adjusted likelihood of being a student than humanitarian-migrant men. Yet the timing of education access differed somewhat between humanitarian migrants of either gender: the share of humanitarian-migrant men in education decreased with time since arrival, whereas the share of humanitarian-migrant women in education increased with time since arrival. The absence of a "female penalty" in relation to education access amongst humanitarian migrants in Australia constitutes a surprising finding.

Altogether, our findings confirmed that humanitarian migrants experience barriers to accessing the Australian higher-education system, and that policy attention is required to redress this situation. However, they also stress that a "one size fits all" policy strategy may be neither sufficient nor appropriate to boost their education prospects. Instead, certain subpopulations within the broader humanitarian-migrant population require special attention from equity practitioners and policymakers—for example, those with low levels of English-language proficiency and those entering the country with low/no educational qualifications.

## **ABOUT THE AUTHORS**

**Francisco (Paco) Perales** is Associate Professor in Sociology at the School of Social Science at The University of Queensland. His research uses longitudinal and life-course approaches to enhance our understandings of socio-economic inequalities in contemporary societies. His recent work has been published in outlets such as *Demography*, *Child Development*, *Social Forces*, *Journal of Marriage and Family*, *Population and Development Review* and *Social Science & Medicine*. Email: f.perales@uq.edu.au

Ning Xiang is a Research Assistant at the Institute for Social Science Research at The University of Queensland. She holds a PhD in social psychology from The University of Queensland and has worked on the national evaluation of the Australian Government's Paid Parental Leave scheme and the follow-up Australian Research Council Linkage project, Millennium Mums Survey. Her current research interests include student engagement, equity in education, gender equity in domestic work, and maternal employment.

Email: n.xiang@uq.edu.au

Lisa Hartley is a Senior Lecturer at the Centre for Human Rights Education at Curtin University. Lisa's interdisciplinary research is focused on questions of human rights and social change which cut across the fields of refugee and migrant studies, sociology, and community and social psychology. She has extensive experience working with a range of community groups providing support and advocacy for human rights issues, informed by a commitment to working along-side communities, including people seeking asylum. Email: Lisa.Hartley@curtin.edu.au

**Matthias Kubler** is a Research Manager at the Institute for Social Science Research and Research Fellow at the ARC Centre of Excellence for Children and Families over the Life Course. He has worked on social research projects in academia and the Queensland public service over the past 16 years. His work over recent years has been focused on policy relevant applications of quantitative research methods in the areas of student retention, equity in Higher Education and graduate outcomes. Email: m.kubler@uq.edu.au **Wojtek Tomaszewski** is Associate Professor at the Institute for Social Science Research. Wojtek has a strong research interest in the impact of disadvantage on educational and labour market outcomes in young people. He has undertaken a number of research projects for various departments in the British government, as well as for both State and Commonwealth Government in Australia. He has published in a number of high-profile academic journals, including *Research in Higher Education, Journal of Youth Studies, Social Indicators Research, Journal of Sociology* and *Social Service Review*.

Email: w.tomaszewski@uq.edu.au

DISCLAIMER: The content of this Working Paper does not necessarily reflect the views and opinions of the Life Course Centre. Responsibility for any information and views expressed in this Working Paper lies entirely with the author(s).

## ABSTRACT

Humanitarian migrants are amongst the most marginalised population groups in countries within the Global North, including Australia. An important channel for these migrants to successfully settle into the host society and improve their socio-economic outcomes is participation in the local education system, particularly in higher-education options. However, we know surprisingly little about the socio-demographic factors that facilitate or constrain access to (higher) education amongst humanitarian migrants, with evidence from robust quantitative studies being particularly scarce. The present study fills this important gap in knowledge by analysing Australian longitudinal survey data (Building a New Life in Australia; n=2,109 migrants and 8,668 person-year observations) by means of random-effect panel regression models. Key results indicated that higher Englishlanguage proficiency and pre-arrival education levels are core factors fostering greater engagement with the Australian higher-education system amongst humanitarian migrants. Surprisingly, humanitarian-migrant women in our sample exhibited a greater adjusted likelihood of being a student than humanitarian-migrant men. Altogether, our findings confirmed that humanitarian migrants experience barriers to accessing the Australian higher-education system, and that policy attention is required to redress this situation. However, they also stress that a "one size fits all" policy strategy may be neither sufficient nor appropriate to boost their education prospects.

Keywords: Australia; education; equity; humanitarian migrants; refugees; university

**Suggested citation:** Perales, F., Xiang, N., Hartley, L., Kubler, M., & Tomaszewski, W. (2020). 'Understanding access to higher education amongst humanitarian migrants: A longitudinal analysis of Australian survey data'. Life Course Centre Working Paper Series, 2020-31. Institute for Social Science Research, The University of Queensland.

#### 1. Introduction

The global population of forcibly displaced individuals stands at a historical high, at approximately 70.8 million (UNHCR, 2018). Of these, ~20.4 million people are formally recognised as refugees under the UNHCR mandate.<sup>1</sup> In Australia, where this study is based, more than 145,000 humanitarian migrants entered the country over the 2006-2016 period (Parliament of Australia, 2017). Despite the support offered by government institutions and third-sector organizations, humanitarian migrants remain amongst the most marginalised population groups in countries across the Global North (Baum, Lööf, Stephan, & Zimmermann, 2020; OECD, 2016, 2019a, 2019b), including Australia (e.g., Hugo, 2011; Szkudlarek, 2019). For example, the employment rates of recently arrived working-age humanitarian migrants in Australia, 33.3% for men and 7.5% for women (DSS, 2017), are strikingly lower than those of the Australian-born population, 81.1% for men and 69.6% for women (Wilkins & Lass, 2018). This is consistent with global evidence showing that many refugees around the world are unemployed, under-employed or work in jobs that do not match their skills (Szkudlarek, 2019).

An important channel to enable humanitarian migrants to settle into the host society and improve their socio-economic outcomes is participation in the local education system. Attainment of home-country educational credentials—particularly higher-education (HE) credentials—not only creates employment pathways for humanitarian migrants; it also improves their cultural-competency skills, widens their social networks, and enhances their subjective wellbeing (Cerna, 2019; Streitwieser, Loo, Ohorodnik, & Jeong, 2018; UNHCR, 2019). However, humanitarian migrants face multiple and unique barriers to accessing and participating in the host-country HE system, both internationally (e.g., Bajwa et al. 2017; Brücker et al., 2016; Cerna, 2019) and in Australia (Baker, Ramsay & Lenette, 2019; Correa-Velez, Barnett & Gifford, 2015; Harris & Marlowe, 2011; Hartley et al., 2018). For these reasons, scholars and policymakers have called for further support for this group to engage

<sup>&</sup>lt;sup>1</sup> According to the *1951 United Nations Convention on the Status of Refugees* and its 1967 Protocol, a refugee is any person who owing to a well-founded fear of being persecuted for reasons of race, religion, nationality, membership of a particular social group or political opinion, is outside the country of his/her nationality and is unable, or owing to such fear, unwilling to avail themselves to the protection of that country. A person seeking asylum, or asylum seeker, is someone who has sought protection as a refugee, but whose claim for refugee protection has not yet been finalized. "Humanitarian migrants" denotes *"people who have successfully applied for asylum and have been granted some sort of protection - refugee or other status"* (OECD, 2016, p.7). When reviewing the available academic evidence, this study considers studies focusing on refugees, humanitarian migrants, and/or asylum seekers—noting that these definitions are not always employed consistently across studies. The empirical analyses presented here will focus on the category of humanitarian migrants—as formally defined by the protection visas offered as part of Australia's Refugee and Humanitarian Program.

in HE (Bajwa et al. 2017; Baker et al., 2019; Brücker et al., 2016; Cerna, 2019; Delaporte & Piracha, 2018; Hartley et al., 2018; Jungblut, Vukasovic, & Steinhardt, 2020).

Policy responses aimed at lifting the educational engagement of humanitarian migrants should be based on a robust body of empirical evidence. While qualitative research has provided valuable insights into the experiences of humanitarian migrants in the hostsociety's education system, quantitative studies on this topic remain scarce (Ben-Moshe, Bertone, & Grossman, 2008; Cerna, 2019; Mestan, 2016; Naidoo, Wilkinson, Adoniou, & Langat, 2018; Ramsay & Baker, 2019; Sladek & King, 2016; Terry et al., 2016; Streitwieser et al., 2018). Generalizable, quantitative evidence on the associations between humanitarian migrants' circumstances and access to HE would further help guide and develop programs and interventions aimed at improving their educational engagement. Because of the greater economic and social rewards that stem from participation in HE (see e.g. ABS, 2017; Desjardins & Lee, 2016; Heckman, Humphries & Veramendi, 2016), it is particularly significant for new studies to consider barriers to accessing HE. This study fills this gap in knowledge. Specifically, it presents the results of analyses of longitudinal survey data examining the socio-demographic factors that facilitate or constrain access to education and HE amongst a recent cohort of humanitarian migrants to Australia. The analyses consider also differences between men and women, thereby recognising the potential gendering of these processes (Harris, Spark, & Ngum Chi Watts, 2015; Hatoss & Huijser, 2010; Seck, 2015).

#### 2. Literature review

#### 2.1 Barriers to education amongst humanitarian migrants

Despite recent efforts to increase humanitarian migrants' education access and participation in countries such as Australia, Sweden, the US, the UK and Germany (e.g., Jungblut et al., 2020; Mangan & Winter, 2017; Manhica, Berg, Almquist, Rostila, & Hjern, 2019; Morrice, 2009; Ramsay & Baker, 2019), international evidence suggests that the educational status of this population remains a matter of concern (Streitwieser et al., 2018; UNHCR, 2019). For example, globally, only ~63% of refugee children attended primary school in 2018 and just ~24% of refugee adolescents attended secondary school (UNHCR, 2019). The picture is particularly bleak for HE, with only ~3% of eligible refugee adolescents being enrolled in HE in 2018, compared to ~37% of eligible non-refugee adolescents. Recognising the crucial role of education in the successful resettlement of refugees, the UNHCR set up the ambitious target "to achieve enrolment of 15% of college-eligible refugees in tertiary or connected higher education programmes in host and third countries" (UNHCR, 2019, p.13).

Much of the literature examining barriers to education amongst humanitarian migrants has focused on access to and success in primary and secondary education (Streitwieser et al., 2018). However, a growing body of studies has explored refugee access to and participation in HE (see Ramsay & Baker, 2019 for a review). Previous research has identified several barriers to accessing HE amongst humanitarian migrants (Bajwa et al., 2017; Hatoss & Huijser, 2010; Streitwieser et al., 2018; UNHCR, 2019). These can be categorised into: material/financial barriers, health barriers, informational barriers and gendered barriers.

Material and financial barriers refer to issues such as the lack of financial resources to pay for tuition, materials or transportation and other education-related fees (UNHCR, 2019). Financial barriers may also emerge when refugee students need to prioritise family financial needs over HE-for instance, when they have to work to support their family instead of undertaking HE studies (UNHCR, 2019). Health barriers refer to the fact that refugees often experience health *sequelae* stemming from traumatic experiences of persecution, violence or retention-including mental disorders such as post-traumatic stress disorder (see e.g., Ibrahim & Hassan, 2017). In addition, refugee adolescents may exhibit delays in their cognitive and socio-emotional developmental trajectories due to spending long portions of their childhood in refugee camps (Cerna, 2019). Informational barriers refer to difficulties with accessing professional support and navigating educational pathways (Bajwa et al., 2017; Morrice, 2009). For example, Morrice (2009) documents that individuals from refugee backgrounds find it hard to access accurate information that helps them navigate the UK educational system. Similarly, Bajwa et al. (2017) found that refugee students in Canada struggled to find useful information on how to have their credentials assessed, finance their HE studies, or utilise HE online resources.

Gendered barriers refer to the challenges that some refugee women may face in accessing HE and which stem from particular cultural gender-role practices and expectations (e.g., Harris et al., 2015; Hatoss & Huijser, 2010; Seck, 2015). For example, drawing on interviews with 14 Sudanese refugees who had recently arrived in Australia, Hatoss and Huijser (2010) documented that cultural-gender factors played a key role in restricting opportunities for women to pursue HE by locating women's roles within the home. Similar gendered barriers have been documented elsewhere, such as Karen refugee women in Australia (Watkins, Razee & Richters, 2012).

When considering the factors that constrict refugees' access to HE, it is important to avoid deficit perspectives that treat these factors as attributes inherent to the individual. Some studies frame these barriers in terms of the support that institutions provide for refugee students in pursuing HE or as the product of structural inequalities (e.g., Naylor, Terry, Rizzo, Nguyen, & Mifsud, 2019). For example, Jungblut et al. (2020) identified three main institutional barriers to access HE faced by refugees in Germany–namely, the scarcity of preparatory German-language courses, the absence of a central education-credential recognition authority and humanitarian migrants' inability to provide official documents that prove their course eligibility (e.g., previously earned diplomas).

#### 2.2 The Australian context and emerging evidence base

Australia is a traditional destination country for migrants from across the globe and currently maintains multiple schemes or programs that guide its immigration intake. This includes a Refugee and Humanitarian Program aimed at helping people in humanitarian need who are (i) outside Australia ("offshore") and need to resettle to Australia, as well as (ii) people in humanitarian need who are already in Australia ("onshore") and seek protection after arriving in Australia. Most humanitarian visas to Australia are granted through the Offshore Humanitarian Program. Over 17,000 humanitarian visas were granted under the Australian Government's humanitarian migration program during 2018-2019 (DHA, 2019). In the 2018-2019 period, the largest number of such visas was granted to people born in Iraq (41.5%), Congo (12.4%), Myanmar (11.7%), Syria (10.7%) and Afghanistan (7.7%) (DHA, 2019).

A mature body of Australian research has deployed *qualitative* methods to understand barriers to humanitarian migrants' access to and success in HE (e.g., Baker, Irwin et al., 2019; Earnest, Joyce, De Mori, & Silvagni, 2010; Harris et al., 2015; Harris & Marlowe, 2011; Hartley et al. 2018; Hatoss & Huijser, 2010; Joyce, Earnest, De Mori, & Silvagni, 2010; Kong et al., 2016; Naidoo, 2015). Consistent with the international literature, these studies also point out that humanitarian migrants face barriers to accessing and participating in the HE system, including strenuous family and/or financial responsibilities while undertaking their studies (e.g., supporting family in Australia or back home) (Joyce et al., 2010), challenges adapting to new educational contexts and community expectations (Harris & Marlowe, 2011) and inadequate HE support systems that fail to recognise refugee's unique needs (Earnest et al., 2010). However, these studies tend to focus on the experiences of humanitarian migrants that have already entered the Australia HE system, rather than on barriers to entry. As an exception, Stevenson and Baker (2018) discuss barriers to HE for humanitarian-migrant students along three stages of student life cycle: HE access, HE participation, and transitions

out of HE. Some core barriers, such as interrupted educational experiences and past and ongoing trauma, have negative impacts throughout all three stages. Yet other barriers are particularly relevant for HE access, such as low school performance, lack of English-language proficiency, lack of access to information on HE pathways, and absence of targeted support (Stevenson & Baker, 2018).

In contrast to the well-developed qualitative body of evidence, only one *quantitative* study to date has examined access to HE amongst humanitarian migrants in Australia. Specifically, Terry et al. (2016) analysed 2009-2014 administrative data from the HE Information Management System (HEIMS) to draw a picture of refugee students' HE participation. Their findings indicated that the number of students from humanitarian-migrant backgrounds enrolling in Australian HE increased from 1,687 in 2009 to 3,506 in 2014 (which corresponds to between 0.21% and 0.34% of all domestic students). While the percentage of refugee-women students also increased from 30% to about 40% of all refugee students, this share varied markedly by country of origin—ranging from 22.5% for Bhutan to 51.3% for Iraq. Their results also indicated that, in 2014, the majority of refugee students were mature aged, with only ~12% of them being less than 20 years old and approximately half (48.7%) of them being 26 years old, or older. This study, however, did not engage with *barriers* to accessing HE.

#### 2.3 The current study: Aims and contributions

This study makes a significant contribution to the literature regarding the factors that facilitate access to education and HE amongst humanitarian migrants by redressing several of its current shortcomings. As Cerna (2019) noted, globally, research on this area remains *"rather limited, fragmented and case specific"* (p.4).

A first limitation of the available evidence base is that the vast majority of studies follows a qualitative methodological approach, both internationally and in Australia (Cerna, 2019; Ramsay & Baker, 2019). Ramsay and Baker (2019) reviewed 46 journal articles and noted "*a strong commitment to qualitative inquiry in the field*" (p.79). Specifically, all 32 *empirical* articles included in the review relied on qualitative methods. While some studies collected data via mixed methods (typically a small-scale survey in conjunction with individual interviews or focus groups), the quantitative data collected in these studies were not representative and rarely analysed using rigorous statistical techniques (Ramsay & Baker, 2019). The present study contributes to the evidence base by undertaking robust quantitative analyses of a large-scale survey, providing results that can be generalised to a recent cohort of humanitarian migrants to Australia. Further, the richness of the survey data at hand allows us to examine the role of a variety of individual- and family-level factors, and to offer novel insights into humanitarian migrants' educational trajectories. We also take advantage of the data to pay attention to the potentially distinct ways in which humanitarian migrant men and women access education and HE.

Second, most of the available evidence concerning humanitarian migrants' interactions with the HE system has focused on humanitarian migrants who had already been admitted into universities, both internationally (e.g., Cerna, 2019) and in Australia (e.g., Baker, Irwin et al., 2019; Joyce et al., 2010; Kong et al., 2016; Naidoo, 2019; O'Rourke, 2011). Hence, their findings are most relevant to identifying barriers to success in—rather than access to—HE. The present study will fill this gap in knowledge by considering the socio-demographic factors of humanitarian migrants that are associated with increasing or decreasing odds of accessing education, and HE more specifically.

#### 3. Data and methods

#### 3.1 Dataset and sample

Building a New Life in Australia: The Longitudinal Study of Humanitarian Migrants (BNLA) is an internationally distinctive, longitudinal study of humanitarian migrants (Edwards, Smart, De Maio, Silbert, & Jenkinson, 2017). The study has interviewed a sample of 2,399 humanitarian migrants from 1,510 households within Australia on an annual basis between 2013/2014 (Wave 1) and 2017/2018 (Wave 5). The in-scope population for the BNLA study comprises adult humanitarian migrants settling in Australia with a permanent visa between May and October 2013, with the sample selected using complex probabilistic methods (AIFS, 2018). The study collects information from two types of humanitarian migrants: (i) "offshore migrants" who received a permanent humanitarian visa overseas and arrived in Australia between May 2013 and December 2013 and (ii) "onshore migrants" who sought asylum after arriving in Australia and were subsequently granted a permanent humanitarian visa between May 2013 and December 2013 (AIFS, 2018, p.3). All survey materials were translated into the respondents' mother-tongue or preferred languages and bilingual interviewers and interpreters were engaged as required. The study's initial response rate was ~55%, with the subsequent wave-on-wave response rates being ~80%. Our analyses of BNLA data are based on the subsample of responding individuals of working age (18 to 64 years), encompassing 8,668 observations from 2,109 individuals.

#### 3.2 Measures

The outcome variables of interest are two binary variables capturing whether at the time of the interview the respondent was studying a course at an Australian educational institution (1=yes, 0=no). We consider both: (i) taking part in *any* type of course and (ii) taking part in a HE course. The latter involves differentiating between University degrees (1=yes) and trade or technical qualifications, paid traineeships, work experience, secondary school (grades 7-12) and short courses (0=no). The explanatory variables capture a range of factors known to affect individuals' decisions concerning their economic and education participation. These include continuous measures of respondent's age (expressed in years) and its square (to capture non-linear relationships), self-assessed spoken-English proficiency, self-assessed general health and number of children in the household; binary variables capturing whether the respondent is a woman, migrant type (onshore/offshore), area of residence (regional/non-regional), current employment status, and being ever in paid work prior to arriving in Australia; and sets of dummy variables capturing respondent's highest educational qualification prior to arriving in Australia, partnership status, length of time in Australia; and country of birth. Table 1 presents means and standard deviations for all analytic variables, pooling observations from survey participants across time points.

	Mean/%	SD
Currently studying in Australia (%)		
Any course	12.6	
Degree course	1.7	
Trade/technical course, or a paid traineeship	5.0	
Some other course, or work experience	5.3	
Woman (%)	45.3	
Age	37.3	11.6
Spoken English proficiency (1-4)	2.3	0.8
General health (1-6)	3.9	1.4
Level of education pre-arrival (%)		
None	15.2	
Some schooling	67.8	
Trade gualification	6.1	
Degree gualification	10.8	
Marital status (%)		
Married, partnered	62.6	
Divorced, separate or widowed	10.0	
Single, never married	27.4	
Number of children in the household	1.1	1.2
Onshore migrant (%)	14.7	
Lives in a regional area (%)	8.5	
Currently in paid employment (%)	20.6	
Ever in paid employment pre-arrival (%)	56.0	
Time in Australia (%)		
<1 year	21.0	
1 year	20.1	
2 years	18.8	
3 years	19.1	
4+ years	21.0	
Country of origin (%)		
Iraq	41.7	
Afghanistan	25.6	
Iran	11.1	
Mvanmar	5.8	
Other country	1.1	
n (observations)	-	8,668
n (groups)		2,109

 Table 1. Sample means and standard deviations for analytic variables

Notes: BNLA, Waves 1-5 (2013-2017). SD: Standard deviation.

#### 3.4 Analytic approach

Because BNLA is a panel dataset, we fit random-effect logistic regression models that take into account the nesting of observations within the same individuals over time. These models estimate the coefficients on the explanatory variables using a weighted average of the between and within effects (Wooldridge, 2010). The analyses use random-effect *logistic* regression models because the outcomes capturing access to education are all dichotomous variables. Formally, the models take the following form:

$$log\left(\frac{pr(E_{it}=1)}{1-pr(E_{it}=1)}\right) = \alpha + (\beta S_{it}) + u_i$$
(1)

where subscripts *i* and *t* stand for individual and time period, respectively; *E* is a binary measure capturing being enrolled in an education or HE course in Australia; S represents the set of socio-demographic variables introduced before; *a* is a model intercept; *B* is the vector of coefficients of interest to be estimated; and *u* is a person-specific random intercept (i.e., a random effect). For ease of interpretation, model coefficients are expressed as odds ratios (ORs). We subsequently examine gender differences in access to education by fitting analogous models in which all of the explanatory variables were interacted with a "woman" dummy variable.

#### 4. Empirical evidence

#### 4.1 Levels of and trends in educational engagement

The top of Table 1 presents sample averages for the educational variables of interest. Pooling all time periods, 12.6% of humanitarian migrants were studying towards a qualification in Australia; 1.7% were studying towards a degree, 5% a trade/technical course or a paid traineeship, and a further 5.3% another course or work-experience program.<sup>2</sup> That is, around 14.2% of all *students* were studying towards a degree. These figures pertain to the overall sample and do not yield any insights into temporal trends. The results presented in Table 2, however, are split by survey wave. The pooled figures in Panel A reveal that the percentage of humanitarian migrants who were studying any course increases between Wave 1 (10.5%) and Wave 3 (15.4%), but decreased thereafter (11.2% in Wave 5). The percentage

 $<sup>^{2}</sup>$  The percentages for the different education categories (in this case, 5.3+5.0+1.7) do not add up to the overall percentage (in this case, 12.6) because a small number of respondents reported undertaking multiple courses falling into more than one of the categories. This also applies to other education variables.

of humanitarian migrants who studied towards a degree, however, increased linearly over time: from 0.5% in Wave 1 to 2.6% in Wave 5.

The second and third panels present the figures for men (Panel B) and women (Panel C) separately. Across survey waves, more humanitarian-migrant women (14.3%) than men (11.2%) were current students, with some evidence of gender differences in time trends. For example, the share of women who were current students raised from 10.7% in Wave 1 to 14.4% in Wave 5, while a decrease was observed for men (from 10.3% to 8.4%). The increase in the share of women accessing HE (0.8% in Wave 1 and 3.6% in Wave 5) was also much more pronounced than the equivalent increases for men (0.4% in Wave 1 and 1.7% in Wave 5).

	_	Mean/%			
	Wave 1	Wave 2	Wave 3	Wave 4	Wave 5
Panel A: All respondents					
Any course	10.5	14.1	15.4	12.4	11.2
Degree course	0.5	1.2	2.2	2.5	2.6
n (observations)	2,079	1,695	1,585	1,609	1,568
Panel B: Men					
Any course	10.3	13.4	13.0	10.9	8.4
Degree course	0.4	1.2	2.1	1.9	1.7
n (observations)	1,146	954	865	875	844
Panel C: Women					
Any course	10.7	15.0	18.2	14.3	14.4
Degree course	0.8	1.3	2.2	3.1	3.6
n (observations)	933	741	720	734	724

 Table 2. Sample means for outcome variables, by survey wave

Notes: BNLA, Waves 1-5 (2013-2017).

#### 4.2 Predictors of engagement in education

The results of random-effect logistic regression models examining the socio-demographic factors associated with being a current student in Australia are shown in columns (1) and (2) in Table 3. The results for being a current student in any educational stream are presented in Column (1). All else being equal, the following factors significantly increased the odds of being a current student in Australia: being a woman (OR=1.54, p<0.01); higher levels of spoken English-language proficiency (OR=2.43, p<0.01); better health (OR=1.12, p<0.01); being single, rather than married/partnered (OR=2.43, p<0.01); and coming from Afghanistan (OR=1.44, p<0.05), Iran (OR=2.48, p<0.01), Myanmar (OR=5.22, p<0.01) or another country (OR=1.76, p<0.01), rather than from Iraq. Meanwhile, other factors significantly decreased the odds of being a current student, *ceteris paribus*: having already

completed an Australian qualification (OR=0.18, p<0.01); having no educational qualifications (OR=0.36, p<0.01), some schooling (OR=0.46, p<0.01) or a trade qualification (OR=0.63, p<0.1), compared to a degree, prior to arriving in Australia; being an onshore rather than an offshore migrant (OR=0.59, p<0.01); undertaking paid employment (OR=0.68, p<0.01); and having been in Australia less than 1 year (OR=0.45, p<0.01) or 1 year (OR=0.74, p<0.05), compared to 2 years.

	Currently studying		
	Any course	Degree <sup>a</sup>	
	(1)	(2)	
Already completed Australian qualification	0.18***	3.26 <sup>*</sup>	
Respondent is a woman	1.54***	2.62*	
Age	1.03	0.85	
Age squared	1.00	1.00	
Spoken English proficiency (1-4)	2.43***	4.46***	
General health (1-6)	1.12***	1.33	
Level of education pre-arrival (ref. Degree			
qualification)			
None	0.36***	0.01**	
Some schooling	0.46***	0.09***	
Trade qualification	0.63*	0.21	
Marital status (ref. Married/Partnered)			
Divorced, separate or widowed	1.12	2.11	
Single, never married	2.43***	2.92	
Number of children in the household	1.01	0.71	
Onshore migrant	0.59***	1.03	
Lives in a regional area	1.32	2.50	
Currently in paid employment	0.68***	0.85	
Ever in paid employment pre-arrival	0.84	2.67*	
Time in Australia (ref. 2 years)			
<1 year	0.45***	0.41	
1 year	0.74**	0.63	
3 years	0.85	5.58***	
4+ years	0.84	8.61***	
Country of origin (ref. Iraq)			
Afghanistan	1.44**	2.09	
Iran	2.48***	3.14	
Myanmar	5.22***	0.07	
Other country	1.76***	2.30	
n (observations)	8,492	1,070	
n (groups)	2,109	668	
Pseudo R <sup>2</sup>	0.04	0.12	

 Table 3. Odds ratios from random-effect logistic regression models

*Notes*: BNLA, Waves 1-5 (2013-2017). <sup>a</sup> Conditional on studying for a qualification of any kind. Statistical significance: p<0.1, p<0.05, p<0.01.

The magnitude of these associations is better grasped by visual inspection of the left panel of Figure 1, in which the model results are transformed into predicted probabilities. This hints at which of the relationships of interest are substantially—as opposed to statistically— significant. As can be appreciated in the figure, the differences in humanitarian migrants' access to the Australian education system were greatest across the categories of pre-arrival education, country of birth, English-language proficiency and—to a lesser extent—general health. Take for example the observed differences by the educational qualifications humanitarian migrants held at the point of arriving in Australia. Adjusting for the model controls, those who entered Australia with no qualifications participated in the education system in 10.1% of the observations. The analogous figures are visibly higher amongst those who entered Australia with some schooling (11.9%), a trade qualification (14.5%) and—particularly—a university degree (19%). The gradient for English-language proficiency is equally striking: respondents who reported speaking "not at all" English had a 5% probability of being a student, compared to a much higher 28.5% amongst respondents who reported speaking English "very well".



Figure 1. Predicted probabilities from random-effect logistic regression models of studying towards an Australian qualification & studying towards a degree

Notes: BNLA, Waves 1-5 (2013-2017). Based on results from the model presented in Column (1) in Table 3 and Column (2) in Table 3. For discrete variables (e.g., marital status), we present predicted probabilities for each of the variable's categories. For continuous variables (e.g., age), we selected representative values of the variable's distribution and presented probabilities for those values.

#### 4.3 Predictors of higher-education access

Column (2) in Table 3 shows the results of a model examining which characteristics were associated with individuals studying towards a degree rather than other courses, out of those individuals who were studying towards any type of qualification. The model coefficients indicated that such odds were significantly higher amongst women than men (OR=2.45, p<0.1), and those who had been employed pre-arrival compared to those who had not (OR=2.67, p<0.1), and increased with English-language proficiency (OR=4.46, p<0.01) and time since arrival. The odds of studying towards a degree were, however, negatively associated with having arrived in Australia with no qualifications (OR=0.01, p<0.05) or only some schooling (OR=0.09, p<0.05), compared to having arrived with a university degree. Of note, some of the ORs in the model appear to be large but are not statistically significant. This is likely due to reduced statistical power stemming from the smaller sample size in this model, which is restricted to individuals who were studying for a course.

A visual representation of the magnitude of these results as predicted probabilities is displayed in the right panel of Figure 1. The figure shows that the most pronounced differences were for pre-arrival education, English-language proficiency, time since arrival and—to a lesser extent age and general health. This pattern is therefore similar to the one observed for studying towards any qualification. However, some of these factors appear particularly important for studying towards a university degree, including prior educational qualifications and time since arrival. For instance, differences by educational qualifications pre-arrival indicated that those who entered Australia with no qualifications selected a HE course in 4.1% of observations, compared to 9.4% amongst those who entered with some schooling, 13% amongst those with a trade qualification, and 21.3% amongst those with a university degree. For time since arrival, a substantially larger share of humanitarian migrants opted for HE courses when they had been in Australia for 3 years (16.1%) or 4+ years (18.6%), compared to less than 1 year (5.8%), 1 year (7%) or 2 years (8.5%). Other factors that appear particularly relevant for HE access include having a younger age (20 years or younger) and having no children.

#### 4.4 Gender differences

In a final set of analyses we examined whether there were differences between humanitarianmigrant men and women in the predictors of being a (HE) student.<sup>3</sup> The results of these interactive models—presented in Table A1—are quite extensive and difficult to interpret. For this reason, the discussion here focuses on a selection of key findings; specifically, the results of socio-demographic factors for which statistically significant gender differences were observed. The model results revealed statistically significant gender differences for just a handful of variables: having previously completed an Australian qualification, time spent in Australia, employment status and marital status.<sup>4</sup> A review of the predicted probabilities associated with these parameters (not shown) indicated that the most substantial differences were those pertaining to time in Australia, employment status and marital status. These are discussed in turn, and displayed in Figure 2 as predicted probabilities.

The top panel of Figure 2 demonstrates that both humanitarian-migrant men and women appear to engage with the education system at similar rates in the first two years in the country. However, while women's rates grow over time, men's rates decrease. This may reflect a propensity for humanitarian-migrant women to take the bulk of the domestic responsibilities over the early settlement period (see e.g., Hatoss & Huijser, 2010; Seck, 2015). The middle panel of Figure 2 shows how paid employment seems to be somewhat of a barrier to accessing education amongst humanitarian-migrant men, but markedly increases the probability of accessing the education system among humanitarian-migrant women. It is possible that men and women are opting for different types of qualifications, with different purposes: women may target qualifications to upskill while already in a job, while men may target qualifications that improve their chances of moving quickly into employment. Either way, this pattern of results indicates that a greater share of humanitarian-migrant women than men juggle work and study.

<sup>&</sup>lt;sup>3</sup> Due to small cell sizes when disaggregating the sample by gender, it was not possible to undertake analyses of whether the qualifications that humanitarian migrants studied for or had attained were university degrees or other type of courses.

<sup>&</sup>lt;sup>4</sup> A possible explanation is driven by the properties of the data. The statistical interactions effectively split the BNLA sample between respondents who are men and women, thereby halving the statistical power to detect statistically significant differences. For a few of the variables across the models, however, there were *statistically* significant gender disparities. However, these differences were seldom *substantially* significant—that is, their magnitude was rarely large.

Finally, the bottom panel of Figure 2 makes it apparent that being single is a significant incentive for humanitarian-migrant women but not men to become a student. This may reflect a reliance on a male-breadwinner model amongst humanitarian-migrant women, with those who are single being incentivised to upskill and, in doing so, augment their labour-market prospects.



Figure 2. Predicted probabilities from random-effect logistic regression models of the odds of being a student, by gender and time in Australia, employment status and marital status

Notes: BNLA, Waves 1-5 (2013-2017). Based on results from the model presented in Table A1.

Single

0

MarriedPart

Div.ISep. Mid.

Marital status

MarreolPart

Div.1580 Mid.

Single

#### 5. Discussion and conclusion

This study set out to contribute unique empirical evidence on the socio-demographic factors that act as barriers or enablers to humanitarian migrants' accessing education, and HE more specifically, using a unique quantitative dataset (BNLA). The analyses yielded several key findings. First, while only a small share of humanitarian migrants started an educational course early into their settlement period, an upwards trend over the 5-year observation window was observed. By the end of the fifth year, 15.4% of humanitarian migrants were enrolled in a course (2.6% in a HE course). Although this time trend is a reason for moderate optimism, engagement with the HE system amongst humanitarian migrants in Australia is modest: of all students, only 14.2% pursued HE. This evidence supports the notion that this population group does indeed experience barriers to accessing the Australian HE system.

Second, two factors consistently predicted greater engagement with the Australian education system and with higher-order options within it: (i) English-language proficiency and (ii) education level at the time of arrival. In other words, within this cohort of humanitarian migrants, developing English-language proficiency and limited educational experiences acted as key barriers to education/HE in Australia. The importance of other factors that could be perceived as plausible barriers to education/HE amongst humanitarian migrants (e.g., age, general health, marital status, parenthood and onshore migration) was smaller. The critical role of mastery in the English language observed here is, however, not surprising. In fact, improving local-language proficiency has been identified by the OECD as a priority area for the educational integration of refugees (Cerna, 2019, p.34). Cerna (2019) discusses a number of strategies across OECD countries aimed at improving refugees' skills in the host-country language, including introductory/welcome language courses and transition-to-mainstream-language programs. In Australia, the Adult Migrant English Program (AMEP) has historically provided up to 510 hours of English-language tuition to eligible humanitarian migrants to help them learn the foundations of the English language, as well as cultural skills that enable them to participate socially and economically in Australian society (Department of Home Affairs, 2020). Our findings suggest that increasing humanitarian migrants' access to this type of programs and/or the intensity of their exposure may go a long way in enhancing their educational prospects. In this regard, in August 2020, the Australian government made positive changes to the AMEP, including removing the cap of 510 hours and the time limits for enrolment and completion—which previously meant

that humanitarian migrants had to complete the program within 5 years of settlement in Australia (Tudge, 2020).

Similarly, arriving in Australia with degree-level education qualifications was also a factor that consistently predicted engagement in the Australian education system, including HE. This finding is somewhat concerning, as it suggests that those humanitarian migrants with greater need to access the Australian education system (i.e., those with low or no educational credentials at arrival) are those who appear to be least likely to access it. And this finding does not simply reflect that highly-skilled humanitarian-migrants may be more likely to have sufficient educational credentials to be admitted into a HE course. In fact, this group was also more likely than the low-skilled group to access non-HE courses. A possible explanation is that highly educated humanitarian migrants may arrive in Australia with advanced skills and an appreciation of the importance of education for improving their socio-economic prospects, but may hold education credentials that are not officially recognised. For this group of humanitarian migrants, access to education (particularly HE) may be a pathway to validate their existing skills with a formal Australian qualification. Altogether, these results underscore the importance of reaching low-skilled humanitarian migrants for any policies or programs aimed at improving humanitarian-migrants' educational trajectories in Australia. Policies aimed at improving the educational engagement of low-skilled humanitarian migrants may focus on developing their HE aspirations, improving their financial capacity to enrol, and developing the skills required for successful participation (e.g., general literacy and numeracy).

Third, the results revealed significant differences in the probability of studying a course by humanitarian-migrants' country of origin. For example, humanitarian migrants from Iraq were generally found to display lower engagement with education than humanitarian migrants coming from other major source countries. It is possible that humanitarian migrants from Iraq experience the most elevated rates of post-traumatic stress disorder, owing to the intensity, duration and contemporaneity of conflict in their country (Slewa-Younan et al., 2015). Indeed, a wealth of psychological literature documents the long-term consequences of trauma amongst refugee populations post resettlement (Ibrahim & Hassan, 2017). This finding also echoes previous research highlighting the diversity in humanitarian-migrant experiences (e.g., Cerna, 2019; Naylor et al., 2019) and the ensuing complexity in designing intervention programs (OECD, 2016). The observed pattern of results is clearly a cause for concern, as Iraqi entrants are the largest group of recent humanitarian migrants in Australia—amounting to 41.5% of those who

arrived in the 2018/2019 period (DHA, 2019). It is nevertheless important to acknowledge that we cannot separately examine the outcomes of humanitarian migrants from small source countries, and it is possible that migrants from some of these countries experience more disadvantageous circumstances than Iraqi migrants. Regardless, this finding underscores the importance of considering specific countries of origin when providing targeted support to humanitarian migrants.

An additional set of analyses examined the potential gendering of the processes discussed thus far. Specifically, the analyses tested for gender differences in the factors enabling/constraining access to education within this population. A core finding emerging from these analyses was that-compared to humanitarian-migrant men in Australia-humanitarian-migrant women exhibited a greater adjusted likelihood of being a student. The timing of education access differed somewhat between humanitarian migrants of either gender: the share of humanitarianmigrant men in education decreased with time since arrival, whereas the share of humanitarianmigrant women in education increased with time since arrival. The absence of a "female penalty" in relation to education access amongst humanitarian migrants in Australia constitutes a surprising finding. This is because, compared to humanitarian-migrant men, humanitarianmigrant women are disadvantaged in other life domains, most notably in relation to their labourmarket outcomes (ABS, 2018; Delaporte & Piracha, 2018; OECD, 2019b). Similarly, while it is theoretically plausible that humanitarian-migrant women experienced different barriers to accessing education than humanitarian-migrant men, the BNLA results portrayed a picture of similarity rather than one of difference. There were few statistically significant gender differences in the estimated effect of the socio-demographic predictors on the likelihood of studying.

Despite the importance of our findings, there are also study limitations that must be acknowledged, some of which point to potentially fruitful avenues for further research. First, the humanitarian migrants covered in our data exclude refugees on temporary protection visas and safe haven enterprise visas, who are arguably the most disadvantaged subgroups in relation to accessing the Australian (higher) education system (Hartley et al., 2018). Second, while BNLA offers important advantages in terms of the richness of the data collected and its longitudinal design, it does not enable direct comparisons between humanitarian migrants and other population groups. Third, the data pertain to a single cohort of migrants who arrived at a

specific socio-historical point in time and whose experiences and circumstances may or may not coincide with those of newer cohorts of humanitarian migrants with different characteristics and pre-settlement pathways.

Our findings also point to other areas in which future research is needed. One important avenue for further scrutiny involves gaining a more holistic understanding of humanitarian-migrant experiences after graduation. This study has focused on barriers to accessing education, with a particular focus on HE entry, whereas earlier studies have largely examined the experiences of humanitarian migrants while participating in HE (see e.g., Baker, Irwin, et al., 2019; Joyce et al., 2010; Kong et al., 2016; Naidoo, 2019; O'Rourke, 2011). There is nevertheless little evidence on whether or not humanitarian migrants face difficulties upon completing their HE studies and transitioning into the labour market. Australian evidence for other disadvantaged groups has yielded mixed findings (Tomaszewski et al., 2019). As such, studies answering questions such as "do humanitarian migrants with HE educational credentials attain greater labour-market outcomes than humanitarian-migrants with lower educational credentials?" or "are the benefits of HE comparable for humanitarian migrants and other populations?" are sorely needed (Baker, Due, & Rose, 2019).

Altogether, the findings presented here reveal that specific segments of the humanitarianmigrant population are comparatively less likely to access Australian HE. Hence, they suggest that a "one size fits all" policy strategy may be neither sufficient nor appropriate to boost the chances of humanitarian migrants reaping the benefits of HE in the host society. Instead, certain subpopulations within the broader humanitarian-migrant group require special attention from equity practitioners and policymakers—for example, those with low levels of English-language proficiency and those entering the country with low/no educational qualifications.

#### References

- Australian Bureau of Statistics (ABS). (2017). *Education and work, Australia, May 2017*. Canberra: Australian Bureau of Statistics.
- Australian Bureau of Statistics (ABS). (2018). Understanding Migrant Outcomes Insights from the Australian Census and Migrants Integrated Dataset, Australia, 2016, CAT No. 3417.0. Canberra: Australian Bureau of Statistics. Available online at: <u>https://www.abs.gov.au/AUSSTATS/abs@.nsf/Latestproducts/3417.0Main%20Features2</u> 2016?opendocument&tabname=Summary&prodno=3417.0&issue=2016&num=&view=.
- Australian Institute of Family Studies (AIFS). (2018). *Building a New Life in Australia Data Users Guide Release 4.0.* Victoria: Australian Institute of Family Studies. Available online at: <u>http://www3.aifs.gov.au/bnla/userguide/docs/BNLA%20Data%20Users%20Guide%20Rel</u> ease%204.pdf.
- Bajwa, J. K., Couto, S., Kidd, S., Markoulakis, R., Abai, M., & McKenzie, K. (2017). Refugees, higher education, and informational barriers. *Refuge: Canada's Journal on Refugees/Refuge: revue canadienne sur les réfugiés*, 33(2), 56-65.
- Baker, S., Due, C., & Rose, M. (2019). Transitions from education to employment for culturally and linguistically diverse migrants and refugees in settlement contexts: what do we know? *Studies in Continuing Education*, 1-15. doi:https://doi.org/10.1080/0158037X.2019.1683533.
- Baker, S., Irwin, E., Freeman, H., Nance, S., & Coleman, J. (2018). Building cultural and linguistic bridges: Reflections on a program designed to support adult students from refugee backgrounds' transitions into university. *Journal of Academic Language and Learning*, 12(1), A64-A80.
- Baker, S., Irwin, E., Taiwo, M., Singh, S., Gower, S., & Dantas, J. (2019). Methodological diversity as an asset for transition-focused higher education research with students from refugee backgrounds. *Review of Education*, 7(1), 5-32.
- Baker, S., Ramsay, G., & Lenette, C. (2019). Students from Refugee and Asylum Seeker Backgrounds and Meaningful Participation in Higher Education: From Peripheral to Fundamental Concern. Widening Participation and Lifelong Learning, 21(2), 4-19.
- Baum, C. F., Lööf, H., Stephan, A., Zimmermann, K. F. (2020). Occupational Sorting and Wage Gaps of Refugees, GLO Discussion Paper, No. 562. Essen: Global Labor Organization (GLO).
- Ben-Moshe, D., Bertone, S., & Grossman, M. (2008). Refugee access and participation in tertiary education and training. Melbourne: Institute for Community, Ethnicity and Policy, Victoria University,
- Brücker, H., Rother, N., Schupp, J., von Gostomski, C. B., Böhm, A., Fendel, T., . . . Vallizadeh, E. (2016). Forced migration, arrival in Germany, and first steps toward integration. *DIW Economic Bulletin*, *Deutsches Institut für Wirtschaftsforschung (DIW)*, *Berlin*, 6(48), 541-556.
- Cerna, L. (2019). *Refugee education: Integration models and practices in OECD countries.* Paris: OECD.

- Correa-Velez, I., Barnett, A. G., & Gifford, S. (2015). Working for a better life: Longitudinal evidence on the predictors of employment among recently arrived refugee migrant men living in Australia. *International Migration*, *53*(2), 321-337.
- Delaporte, I. & Piracha, M. (2018). Integration of humanitarian migrants into the host country labour market: evidence from Australia. *Journal of Ethnic and Migration Studies*, 44 (15), 2480-2505.
- Department of Home Affairs (DHA). (2019). *Australia's Offshore Humanitarian Program: 2018-*19. Canberra, ACT: Commonwealth of Australia.
- Desjardins, R., & Lee, J. (2016). Earnings and employment benefits of adult higher education in comparative perspective: Evidence based on the OECD Survey of Adult Skills (PIAAC). Los Angeles: UCLA.
- Department of Social Services (DSS). (2017). Building a New Life in Australia (BNLA): The Longitudinal Study of Humanitarian Migrants — Findings from the first three waves. Canberra: Department of Social Services.
- Earnest, J., Joyce, A., De Mori, G., & Silvagni, G. (2010). Are universities responding to the needs of students from refugee backgrounds? *Australian Journal of Education*, *54*(2), 155-174.
- Edwards, B., Smart, D., De Maio, J., Silbert, M., & Jenkinson, R. (2017). Cohort Profile: Building a New Life in Australia (BNLA): the longitudinal study of humanitarian migrants. *International Journal of Epidemiology*, 47(1), 20-20h.
- Harris, A., Spark, C., & Ngum Chi Watts, M. C. (2015). Gains and losses: African Australian women and higher education. *Journal of Sociology*, *51*(2), 370-384.
- Harris, V., & Marlowe, J. (2011). Hard Yards High Hopes: The educational challenges of African refugee university students in Australia. *International Journal of Teaching, Learning and Higher Education* 23(2), 186-196
- Hartley, L., Fleay, C., Baker, S., Burke, R., & Field, R. (2018). *People seeking asylum in Australia: Access & support in higher education*. Perth, Western Australia: National Centre for Student Equity in Higher Education, Curtin University.
- Hatoss, A., & Huijser, H. (2010). Gendered barriers to educational opportunities: Resettlement of Sudanese refugees in Australia. *Gender Education*, 22(2), 147-160.
- Heckman, J. J., Humphries, J. E., & Veramendi, G. (2016). *Returns to education: The causal effects of education on earnings, health and smoking*. Cambridge, MA: National Bureau of Economic Research.
- Hugo, G. (2011). A significant contribution: the economic, social and civic contributions of first and second generation humanitarian entrants: summary of findings: Commonwealth of Australia, DIAC.
- Ibrahim, H., & Hassan, C. Q. (2017). Post-traumatic stress disorder symptoms resulting from torture and other traumatic events among Syrian Kurdish refugees in Kurdistan Region, Iraq. Frontiers in psychology, 8, 241. doi: 10.3389/fpsyg.2017.00241.
- Joyce, A., Earnest, J., De Mori, G., & Silvagni, G. (2010). The experiences of students from refugee backgrounds at universities in Australia: Reflections on the social, emotional and practical challenges. *Journal of Refugee Studies*, 23(1), 82-97.

- Jungblut, J., Vukasovic, M., & Steinhardt, I. (2020). Higher education policy dynamics in turbulent times-access to higher education for refugees in Europe. *Studies in Higher Education*, 45(2), 327-338.
- Kong, E., Harmsworth, S., Rajaeian, M. M., Parkes, G., Bishop, S., AlMansouri, B., & Lawrence, J. (2016). University Transition Challenges for First Year Domestic CALD Students from Refugee Backgrounds: A Case Study from an Australian Regional University. *Australian Journal of Adult Learning*, 56(2), 170-197.
- Mangan, D., & Winter, L. A. (2017). (In)validation and (mis)recognition in higher education: the experiences of students from refugee backgrounds. *International Journal of Lifelong Education*, 36(4), 486-502.
- Manhica, H., Berg, L., Almquist, Y. B., Rostila, M., & Hjern, A. (2019). Labour market participation among young refugees in Sweden and the potential of education: a national cohort study. *Journal of Youth Studies*, 22(4), 533-550.
- Mestan, K. (2016). Access, achievement and outcomes among students from non-English speaking backgrounds. In A. Harvey, C. Burnheim, & M. Brett (Eds.), *Student equity in Australian higher education* (pp. 125-141): Springer.
- Morrice, L. (2009). Journeys into higher education: The case of refugees in the UK. *Teaching in Higher Education*, *14*(6), 661-672.
- Naidoo, L. (2015). Educating refugee-background students in Australian schools and universities. *Intercultural Education*, 26(3), 210-217.
- Naidoo, L. (2019). Traversing the terrain of higher education: experiences of refugee youth on the inside. *International Journal of Inclusive Education*, 1-14.
- Naidoo, L., Wilkinson, J., Adoniou, M., & Langat, K. (2018). *Refugee background students transitioning into higher education: Navigating complex spaces.* Singapore: Springer.
- Naylor, R., Terry, L., Rizzo, A., Nguyen, N., & Mifsud, N. (2019). Structural Inequality in Refugee Participation in Higher Education. *Journal of Refugee Studies, fez077*.
- O'Rourke, D. (2011). Closing pathways: refugee-background students and tertiary education. *Kotuitui: New Zealand Journal of Social Sciences Online*, 6(1-2), 26-36.
- Organisation for Economic Co-operation and Development (OECD). (2016). *Making Integration Work: Refugees and others in need of protection*. Paris: OECD Publishing.
- Organisation for Economic Co-operation and Development (OECD). (2019a). *OECD reviews of migrant education: The road to integration education and migration*. Paris: OECD Publishing.
- Organisation for Economic Co-operation and Development (OECD). (2019b). International Migration Outlook 2019. Paris: OECD Publishing.
- Parliament of Australia. (2017). Australia's Humanitarian Program: a quick guide to the statistics since 1947. Available online at: <u>https://parlinfo.aph.gov.au/parlInfo/download/library/prspub/3599552/upload\_binary/3599552.pdf</u>
- Ramsay, G., & Baker, S. (2019). Higher education and students from refugee backgrounds: A meta-scoping study. *Refugee Survey Quarterly*, *38*(1), 55-82.

- Seck, M. M. (2015). Female West African immigrants in the United States: Challenges and empowering strategies. *Social Development Issues*, *37*(2), 68-79.
- Sladek, R. M., & King, S. M. (2016). Hidden from view? Bringing refugees to the forefront of equity targets in Australian higher education. *International Studies in Widening Participation*, 3(1), 68-77.
- Slewa-Younan, S., Mond, J. M., Bussion, E., Melkonian, M., Mohammad, Y., Dover, H., . . . Jorm, A. F. (2015). Psychological trauma and help seeking behaviour amongst resettled Iraqi refugees in attending English tuition classes in Australia. *International Journal Mental Health Systems*, 9(5). doi:https://doi.org/10.1186/1752-4458-9-5
- Stevenson, J., & Baker, S. (2018). *Refugees in Higher Education: Debate, Discourse and Practice*: Emerald Publishing Limited.
- Streitwieser, B., Loo, B., Ohorodnik, M., & Jeong, J. (2018). Access to higher education for refugees: A Review of Interventions in North America and Europe. Washington, D. C.: Graduate School of Education & Human Development, the George Washington University.
- Szkudlarek, B. (2019). Engaging business in refugee employment: The employers' perspective. Sydney: the University of Sydney.
- Terry, L., Naylor, R., Nguyen, N., & Rizzo, A. (2016). Not there yet: An investigation into the access and participation of students from humanitarian refugee backgrounds in the Australian Higher Education system. Perth: National Centre for Student Equity in Higher Education.
- Tomaszewski, W., Perales, F., Xiang, N. and Kubler, M. (2019). Beyond Graduation: Socio-Economic Background and Post-University Outcomes of Australian Tertiary-Education Students. *Research in Higher Education* (in press), https://doi.org/10.1007/s11162-019-09578-4
- Tudge, A. (2020). *More class hours available to help migrants learn English*. Available online at: <u>https://minister.homeaffairs.gov.au/alantudge/Pages/More-class-hours-available-to-help-migrants-learn-English.aspx</u>
- United Nations High Commissioner for Refugees (UNHCR). (2018). *Global trends: Forced displacement in 2018*: United Nations High Commissioner for Refugees.
- United Nations High Commissioner for Refugees (UNHCR). (2019). *Refugee education 2030: A strategy for refugee inclusion*: United Nations High Commissioner for Refugees.
- Watkins, P.G., Razee, H., & Richters J. 'I'm Telling You ... The Language Barrier is the Most, the Biggest Challenge': Barriers to Education among Karen Refugee Women in Australia. Australian Journal of Education. 2012;56(2):126-141. doi:10.1177/000494411205600203
- Wilkins, R., & Lass, I. (2018). Household, Income and Labour Dynamics in Australia Survey: selected findings from Waves 1 to 16: the 13th Annual Statistical Report of the HILDA Survey. Melbourne: Melbourne Institute of Applied Economic and Social Research.
- Wooldridge, J. M. (2010). Econometric Analysis of Cross Section and Panel Data. MIT Press.

## Appendix

	Currently studying: Any course <sup>a</sup>
Main effects	
Woman	0.21
Already completed Australian qualification	0.24***
Age	0.99
Age squared	1.00
Spoken English proficiency (1-4)	2.29***
General health (1-6)	1.11**
Education: None	0.42**
Education: Some schooling	0.60**
Education: Trade gualification	0.72
Marital status: Divorced, separate or widowed	1.80
Marital status: Single, never married	1.40
Number of children in the household	1.04
Onshore migrant	0.61**
Lives in a regional area	1.62**
Currently in paid employment	0.55***
Ever in paid employment pre-arrival	0.73*
Time in Australia: <1 year	0.52***
Time in Australia: 1 year	0.93
Time in Australia: 3 years	0.88
Time in Australia: $4$ + years	0.00
Country of origin: Afghanistan	1 35
Country of origin: Iran	2 84***
Country of origin: Myanmar	6 27***
Country of origin: Myaninal	1 77**
Interaction offacts	1.77
Woman interacted with	
Already completed Australian qualification	0.45***
	1 11
Age squared	1.11
Age squared Spaken English proficiency (1, 4)	1.00
Spoken English pronciency (1-4)	1.12
General Nedellin (1-0)	1.01
Education: None	0.95
Education: Some schooling	0.69
Education: Trade qualification	0.85
Marital status: Divorced, separate or widowed	0.71
Marital status: Single, never married	3.15
Number of children in the household	0.91
Unshore migrant	1.44
Lives in a regional area	0.59
Currently in paid employment	2.86
Ever in paid employment pre-arrival	1.18
Time in Australia: <1 year	0.78
Time in Australia: 1 year	0.61
Time in Australia: 3 years	0.96
Time in Australia: 4+ years	1.38
Country of origin: Afghanistan	1.08
Country of origin: Iran	0.66
Country of origin: Myanmar	0.61
Country of origin: Other	0.95
n (observations)	8,492
n (groups)	2,109
Pseudo R <sup>2</sup>	0.04

Table A1. Odds ratios from a random-effect logistic regression models with gender interactions

*Notes*: BNLA, Waves 1-5 (2013-2017). Statistical significance: \* p<0.1, \*\* p<0.05, \*\*\* p<0.01.